ss20-Jun-03
Week 26: (23/06 to 27/06)

SM18 Operation SCHEDULE (Monday Morning) Meeting 

· Test Coordinator this week:   V.Chohan 

· MTA Technician on call:  (16-3602) R.Mompo.

1. Planning for this week : 
· TBA1:   3014 :  Test at 1.9 K (1 run foreseen).
· TBA2:   Preparation of the 2008 and cool down.
· TBC2 :  1026 :  Test at 1.9 K (second run)
·  TBC1 : 1023 will come and be prepared.
2.  Problems from Operation Last week (Stephane, Ewald)
· QH test at 1.9 K : follow the procedures.

· Quench at Minimum energy: different values for each bench!.

· Corrector measurement on TBC2 : problem with the gateway. Not solved for the moment. 

· HV test at 1.9 K on TBC2 : humidity on the current leads.

· AQA problem: LF configuration  file not created sometimes.

· MM problems : Sector 0203 in shaft 21 : one coil is lost.

3. Situation From Cryogenics:  J-P Lamboy
Traffic to be expected at the end of the week : (cluster E commissioning)
4. Other General Issues
